Author's response to reviews

Title: Determinants of the range of drugs prescribed in general practice: a cross-sectional analysis

Authors:

Dinny H. de Bakker (d.debakker@nivel.nl)
Dayline S.V. Coffie (dayline@mail.com)
Eibert R. Heerdink (e.r.heerdink@pharm.uu.nl)
Liset van Dijk (l.vandijk@nivel.nl)
Peter P Groenewegen (p.groenewegen@nivel.nl)

Version: 4 Date: 31 July 2007

Author's response to reviews:

Thank you for, in principle, accepting our paper. In response to your question to clarify the ethical approval of the study we added the following text at the end of the first paragraph of the method section:

'The Second National Survey of General practice was carried out according to Dutch legislation on privacy. The privacy regulation of the study was approved by the Dutch Data Protection Authority'.

We used this text in several other articles with secondary analysis of The Second National Survey'.

Best regards
Dinny de Bakker